Author's response to reviews

Title: Evaluation of the effect of hydroalcoholic leaves extract of indigofera spicata forssk. (fabaceae) on blood glucose level of normal, oral glucose loaded and alloxan induced diabetic rodents

Authors:

Eshetie M Birru (meshetie21@gmail.com)
Mohammedbrhan A Ata (mohammedB18@gmail.com)
Zewdneh Shewamene (zeedshow@gmail.com)

Version: 8 Date: 8 May 2015

Author's response to reviews:

Revisions
@. the last paragraph of the background section is elaborated more to justify the reason why this study is done on animal models
@. the reference citation is also restructured both in the main body and reference section.